Author's response to reviews

Title: The Fitness for the Ageing Brain Study II (FABS II): protocol for a randomized controlled clinical trial evaluating the effect of physical activity on cognitive function in patients with Alzheimer's disease

Authors:

Elizabeth V Cyarto (e.cyarto@nari.unimelb.edu.au)
Kay L Cox (kay.cox@uwa.edu.au)
Osvaldo P Almeida (osvaldo.almeida@uwa.edu.au)
Leon Flicker (leon.flicker@uwa.edu.au)
David Ames (dames@unimelb.edu.au)
Gerard Byrne (gerard.byrne@uq.edu.au)
Keith D Hill (keith.hill@nh.org.au)
Christopher D Beer (christopher.beer@uwa.edu.au)
Dina LoGiudice (dina.logiudice@mh.org.au)
Kana Appadurai (kana_appadurai@health.qld.gov.au)
Muireann Irish (m.irish@neura.com.au)
Emma Renehan (e.renehan@nari.unimelb.edu.au)
Nicola T Lautenschlager (nicolatl@unimelb.edu.au)

Version: 5 Date: 7 December 2010

Author's response to reviews:

I corrected the affiliations for three of the authors. I removed an extra "that" from the first line of paragraph 3 in the Background section. I changed the tense of the verb (from past to future) in the description of the two-minute walking test in the Methods section. To match Figure 1, I added that the intervention group receives educational material to the Behavioral Intervention section. Finally, in the Acknowledgements section, I added the initials of the Chief Investigators listed on the project grant.